Author's response to reviews

Title: Influence network linkages across treatment conditions in a randomized controlled trial of two implementation strategies for scaling up evidence-based practices in public youth-serving systems

Authors:

Lawrence A Palinkas (palinkas@usc.edu)
Ian W Holloway (holloway@publicaffairs.ucla.edu)
Eric Rice (ericr@usc.edu)
C. Hendricks Brown (hendricks.brown@northwestern.edu)
Thomas W Valente (tvalente@usc.edu)
Patricia A Chamberlain (pattic@oslc.org)

Version: 4 Date: 16 September 2013

Author's response to reviews: see over
September 16, 2013

RE: MS #1708727357516378 “Influence network linkages across treatment conditions in randomized controlled trials”

Dear Implementation Science Editorial Team:

Attached please find a revised version of the above referenced manuscript. In accordance with the comments and suggestions provided by the editor, we have made the following revisions:

1. We have replaced “interview guide” with “survey” on p. 8, paragraph 2, last sentence.

We have provided two copies of the revised manuscript, one with markups as requested and one clean copy. In the event the manuscript is accepted for publication, we recommend use of the clean copy.

We look forward to hearing from you at your earliest convenience.

Sincerely yours,

Lawrence A. Palinkas, Ph.D.
Albert G. and Frances Lomas Feldman Professor of Social Policy and Health